Author's response to reviews

Title: Pre-fracture residence, rehabilitation, complications and comorbidities: quantifying the components and determinants of total hospital stay for hip fracture

Authors:

Anthony W Ireland (adje.ireland@bigpond.com)
Patrick J Kelly (p.kelly@sydney.edu.au)
Robert G Cumming (Robert.cumming@sydney.edu.au)

Version: 2 Date: 2 May 2014

Author's response to reviews:

The Editors
BMC Health Services Reasearch
Dear Editors
Re Manuscript 9719594201282841

Revisions:
(1) Title page: Title modified to read "Pre-fracture residence, rehabilitation, complications and comorbidities: quantifying the components and determinants of total hospital stay for hip fracture"
(2) Title page: email address for Anthony W Ireland (corresponding author) confirmed as adje.ireland@bigpond.com
(3) Abstract: redrafted to omit section heading "Design" : this material now incorporated into "Methods"
(4) Section labelled 'acknowledgements" (P15) have now been redrafted as follows:

Competing Interests
AWI is an adviser to the Australian Department of Veterans' Affairs. The authors declare that they have no other competing interests.

Author contributions
AWI: Study concept and design, data linkage, data analyses, literature search, report writing. PJK: Study design, data quality review, statistical analyses, report writing and editing. RGC: Data selection, report writing and editing.

(5) Two superscript markers have been included in the body of Table 2 (P20)